Author's response to reviews

Title: Psychiatric and psychosocial assessment of suicide attempters in Japan: a pilot study at a critical emergency unit in an urban city

Authors:

Tomoki Yamada (tomokin@urahp.yokohama-cu.ac.jp)
Chiaki Kawanishi (chiaki@yokohama-cu.ac.jp)
Hana Hasegawa (hanasato@angel.ne.jp)
Ryoko Sato (rykst@mcn.ne.jp)
Akiko Konishi (akikokanakana@hotmail.com)
Daiji Kato (dkato@yokohama-cu.ac.jp)
Taku Furuno (tfuruno@med.yokohama-cu.ac.jp)
Ikuko Kishida (ikukishida@aol.com)
Toshinari Odawara (odaw1913@med.yokohama-cu.ac.jp)
Mitsugi Sugiyama (sugiyama@urahp.yokohama-cu.ac.jp)
Yoshio Hirayasu (hirayasu@yokohama-cu.ac.jp)

Version: 2 Date: 10 July 2007

Author's response to reviews:

Dear Sir,

This is Dr Kawanishi that submitted the article to BMC Psychiatry (#9356777341409594). We revised our manuscript, and re-submitted. We will be pleased if our revised manuscript is now suitable for publication in BMC Psychiatry.

Sincerely yours,

Chiaki Kawanishi, M.D., Ph.D.
Associate Professor
Department of Psychiatry
Yokohama City University School of Medicine
3-9 Fukuura, Kanazawa-ku, Yokohama 136-0004.
Phone: 81-45-787-2667
Fax: 81-45-783-2540
E-mail: chiaki@yokohama-cu.ac.jp